Author’s response to reviews

Title: A genomic copy number variant analysis implicates the MBD5 and HNRNPU genes in Chinese children with infantile spasms and expands the clinical spectrum of 2q23.1 deletion

Authors:

Xiaonan Du (dresmedu@126.com)
Yu AN (ay.fudan@gmail.com)
Lifei Yu (yulifei7711@163.com)
Renchao Liu (lrc360681@gmail.com)
Yanrong Qin (lrc360681@gmail.com)
Xiaohong Guo (xhguo@shmu.edu.cn)
Daokai Sun (daokaisun@163.com)
Shuizhen Zhou (szzhou@shmu.edu.cn)
Bai-lin Wu (bai-lin.wu@childrens.harvard.edu)
yong-hui Jiang (yong-hui.jiang@duke.edu)
Yi Wang (yiwang@shmu.edu.cn)

Version: 5 Date: 29 April 2014

Author’s response to reviews: see over
Dear Editor

Sorry for several typing mistake, we have corrected in red in the new version.

Sincerely

Wang Yi, MD. PhD.
Professor and Vice President
Division of Neurology
Children's Hospital of Fudan University
399 Wan Yuan Road, Shanghai City 20032, P.R. China
E-mail: yiwang@shmu.edu.cn